Author's response to reviews

Title: A 44-year-old patient with a new-onset seizure disorder after vaccination against Japanese Encephalitis: a case report

Authors:

Christoph Schulze (chrisinro@aol.com)

Version: 3 Date: 29 November 2012

Author's response to reviews: see over
Dear Editor in chief,

I enclose the revised manuscript with the title: “A 44-year-old patient with a new-onset seizure disorder after vaccination against Japanese Encephalitis: a case report” as a case report for consideration in the Journal of medical case reports.

I thank the editor for the helpful comment. I will answer the comment given by the editor. Changes in the manuscript are highlighted. I changed the manuscript and hope that it will be suitable for publication now.

Editors comment
The authors has to clearer draw the connection between the vaccination and the seizure.

Answer to the editor
I thank the editor for the helpful comment. For this, I have been searching new references and included them into the discussion to draw the connection between vaccination and seizure clearer.

I hope the revision makes this case more coherent and plausible to the reader and I hope that it is suitable for publication now.
Sincerely,

Christoph Schulze, MD
First author